Author's response to reviews

Title: Burnout and psychiatric morbidity among medical students entering clinical training: a three year prospective questionnaire and interview-based study

Authors:

Marie E Dahlin (marie.dahlin@ki.se)
Bo Runeson (bo.runeson@ki.se)

Version: 3 Date: 2 April 2007

Author's response to reviews:

Dear Editors,

the following changes have been made in the revised manuscript:

1. Dr Dahlin' middle initial added in the author list.
2. One sentence on informed consent added, page 7, last paragraph.
3. That both authors have read and approved of final manuscript is stated under authors' contributions.
4. Changes were made in Tables 1, 2 and 4. In Table 1, means and standard deviations of "Depressive symptoms" have been corrected. In Table 2, minor corrections following the former have been made in model 1. In Table 4, likewise minor changes have been made in the adjusted model. In Tables 2 and 4, the determination of the gender variable has been changed from "female=1" to "woman=1".
5. References format has been corrected.
6. In addition, minor reformulations have been made and typing errors have been adjusted.

Yours,

Marie Dahlin